Dear Editor

World journal of diabetes,

This is regarding my research article entitled “ID: 83481 - Advanced Glycation End Products signaling, and metabolic complications: Dietary approach” which I received from you for addressing the comments appended by the reviewers and editorial office.

We are extremely thankful to the reviewers and editorial office for going through our manuscript with interest and suggest valuable corrections and modifications. The corrections/suggestions provided by the reviewers and editorial office have been taken positively and incorporated in the manuscript with respect. Incorporation of the corrections/modifications suggested has certainly increased the value of our manuscript. We really appreciate the time and energy they have spent to our manuscript.

We have tried our best in replying the comments and hope that our replies have met your expectations and are satisfactory. We have listed below our reply to each comment which was suggested. Please go through the same and feel free to ask if there is anything we have missed out.

Response to reviewer:

Reviewer #1:

Specific Comments to Authors:

1 Title: partly reflect the main subject. The main part is about sources with poor explanation to the metabolic complication.

Response: Title has been updated and highlight in yellow color

2 Abstract and key words summarize the text.

Response: Abstract and key words has been updated and sequentially made more comprehensive.

3. Introduction should be better composed.

Response: It has been updated
4. Methods. This is narrative review and manuscript lacks the methods. The manuscript lacks BPG's standards.
Response: Method has been incorporated in modified manuscript.

5. Discussion. Must be improved in the terms of cardiac and metabolic complications.
Response: New study has been incorporated and updated.

6. Explain under the tables and figures the abbreviations.
Response: It has been incorporated in modified manuscript.

7. References. there are 168 references, however many of them are quite old. 8. Ethics statements. Not applicable
Response: We have included new recent references in modified manuscript.

Response to reviewer:
Reviewer #2:

**Specific Comments to Authors:**
Specific Comments To Authors: No.

Response to reviewer:
Reviewer #3:

**Specific Comments to Authors:**
Specific Comments To Authors: -The report is important and well written. -The teaching point is practical since the main issues (Advanced glycation end product sources: metabolic disease and cardiac complications) are a frequent combination. -The report includes Key Clinical Message.
Response: Thank you for your time!